Author's response to reviews

Title: Bach Flower Remedies for psychological problems and pain: a systematic review

Authors:

Kylie Thaler (kylie.thaler@donau-uni.ac.at)
Angela Kaminski (angela.kaminski@donau-uni.ac.at)
Andrea Chapman (andrea.chapman@donau-uni.ac.at)
Tessa Langley (tessa.langley@hotmail.com)
Gerald Gartlehner (gerald.gartlehner@donau-uni.ac.at)

Version: 4 Date: 6 April 2009

Author's response to reviews:

We have made the necessary formatting revisions:
- references 22, 23, 26 have volume and page numbers, as discussed with Emma
- tables have no vertical rules
- tables use the same references as the main manuscript
- additional text near figure legend has been removed
- footer has been removed

In addition, we have thoroughly re-checked the manuscript for errors, and made some minor corrections to the text.